\begin{thebibliography}{10}
\providecommand{\url}[1]{\texttt{#1}}
\providecommand{\urlprefix}{\\URL: }
\providecommand{\bibAnnoteFile}[1]{%
  \IfFileExists{#1}{\begin{quotation}\noindent\textsc{Key:} #1\\
  \textsc{Annotation:}\ \input{#1}\end{quotation}}{}}
\providecommand{\bibAnnote}[2]{%
\begin{quotation}\noindent\textsc{Annotation:}\ #2\end{quotation}}

\bibitem{CLAS:2021udy}
N.~Zachariou \emph{et~al.}, \emph{{Double polarisation observable $G$ for
  single pion photoproduction from the proton}}, Phys. Lett. B \textbf{817}
  (2021), p. 136304
\bibAnnoteFile{CLAS:2021udy}

\bibitem{CLAS:2021osv}
U.~Shrestha \emph{et~al.}, \emph{{Differential cross sections for
  $\Lambda(1520)$ using photoproduction at CLAS}}, Phys. Rev. C \textbf{103}
  (2021)~(2), p. 025206
\bibAnnoteFile{CLAS:2021osv}

\bibitem{Hayward:2021psm}
T.~B. Hayward \emph{et~al.}, \emph{{Observation of Beam Spin Asymmetries in the
  Process $ep\rightarrow{e}^{'}{\pi}^{+}{\pi}^{-}X$ with CLAS12}}, Phys. Rev.
  Lett. \textbf{126} (2021), p. 152501
\bibAnnoteFile{Hayward:2021psm}

\bibitem{CLAS:2020ngl}
M.~Carver \emph{et~al.}, \emph{{Photoproduction of the $f_2(1270)$ meson using
  the CLAS detector}}, Phys. Rev. Lett. \textbf{126} (2021)~(8), p. 082002
\bibAnnoteFile{CLAS:2020ngl}

\bibitem{PhysRevC.103.L022201}
S.~Adhikari \emph{et~al.}, \emph{Measurement of beam asymmetry for
  ${\ensuremath{\pi}}^{\ensuremath{-}}{\mathrm{\ensuremath{\Delta}}}^{++}$
  photoproduction on the proton at
  ${E}_{\ensuremath{\gamma}}=8.5\phantom{\rule{0.16em}{0ex}}\mathrm{GeV}$},
  Phys. Rev. C \textbf{103} (2021), p. L022201,
  \urlprefix\url{https://link.aps.org/doi/10.1103/PhysRevC.103.L022201}
\bibAnnoteFile{PhysRevC.103.L022201}

\bibitem{CLAS:2020vpr}
T.~Hu \emph{et~al.}, \emph{{Photoproduction of $\eta$ mesons off the proton for
  $1.2 < E_\gamma < 4.7$ GeV using CLAS at Jefferson Laboratory}}, Phys. Rev. C
  \textbf{102} (2020)~(6), p. 065203
\bibAnnoteFile{CLAS:2020vpr}

\bibitem{ADHIKARI2021164807}
S.~Adhikari \emph{et~al.}, \emph{The GlueX beamline and detector}, Nuclear
  Instruments and Methods in Physics Research Section A: Accelerators,
  Spectrometers, Detectors and Associated Equipment \textbf{987} (2021), p.
  164807,
  \urlprefix\url{https://www.sciencedirect.com/science/article/pii/S0168900220312043}
\bibAnnoteFile{ADHIKARI2021164807}

\bibitem{CLAS:2020rdz}
A.~Celentano \emph{et~al.}, \emph{{First measurement of direct photoproduction
  of the $a_2(1320)^0$ meson on the proton}}, Phys. Rev. C \textbf{102}
  (2020)~(3), p. 032201
\bibAnnoteFile{CLAS:2020rdz}

\bibitem{PhysRevC.101.065206}
S.~Adhikari \emph{et~al.}, \emph{Measurement of the photon beam asymmetry in
  $\vec{\gamma}p\ensuremath{\rightarrow}{K}^{+}{\mathrm{\ensuremath{\Sigma}}}^{0}$
  at ${E}_{\ensuremath{\gamma}}=8.5$ GeV}, Phys. Rev. C \textbf{101} (2020), p.
  065206, \urlprefix\url{https://link.aps.org/doi/10.1103/PhysRevC.101.065206}
\bibAnnoteFile{PhysRevC.101.065206}

\bibitem{CLAS:2020mom}
A.~Schmidt \emph{et~al.}, \emph{{Probing the core of the strong nuclear
  interaction}}, Nature \textbf{578} (2020)~(7796), pp. 540--544
\bibAnnoteFile{CLAS:2020mom}

\bibitem{PhysRevC.100.052201}
S.~Adhikari \emph{et~al.}, \emph{Beam asymmetry $\mathrm{\ensuremath{\Sigma}}$
  for the photoproduction of $\ensuremath{\eta}$ and
  ${\ensuremath{\eta}}^{\ensuremath{'}}$ mesons at
  ${E}_{\ensuremath{\gamma}}=8.8\phantom{\rule{0.28em}{0ex}}\mathrm{GeV}$},
  Phys. Rev. C \textbf{100} (2019), p. 052201,
  \urlprefix\url{https://link.aps.org/doi/10.1103/PhysRevC.100.052201}
\bibAnnoteFile{PhysRevC.100.052201}

\bibitem{PhysRevLett.123.072001}
A.~Ali \emph{et~al.}, \emph{First Measurement of Near-Threshold
  $J/\ensuremath{\psi}$ Exclusive Photoproduction off the Proton}, Phys. Rev.
  Lett. \textbf{123} (2019), p. 072001,
  \urlprefix\url{https://link.aps.org/doi/10.1103/PhysRevLett.123.072001}
\bibAnnoteFile{PhysRevLett.123.072001}

\bibitem{CLAS:2018mmb}
P.~Roy \emph{et~al.}, \emph{{First Measurements of the Double-Polarization
  Observables F , P , and H in \ensuremath{\omega} Photoproduction off
  Transversely Polarized Protons in the N* Resonance Region}}, Phys. Rev. Lett.
  \textbf{122} (2019)~(16), p. 162301
\bibAnnoteFile{CLAS:2018mmb}

\bibitem{CLAS:2018kvn}
J.~T. Goetz \emph{et~al.}, \emph{{Study of $\Xi^*$ Photoproduction from
  Threshold to $W = 3.3$ GeV}}, Phys. Rev. C \textbf{98} (2018)~(6), p. 062201
\bibAnnoteFile{CLAS:2018kvn}

\bibitem{CLAS:2018azo}
S.~Lombardo \emph{et~al.}, \emph{{Photoproduction of $K^+K^-$ meson pairs on
  the proton}}, Phys. Rev. D \textbf{98} (2018)~(5), p. 052009
\bibAnnoteFile{CLAS:2018azo}

\end{thebibliography}


\subsection{Michael R. Dugger}

\noindent{\bf{Professional preparation}}

\begin{table}[ht]
%\vspace{2mm}
\centering
\begin{tabular}{llll}
 &\underline{ {Institution}} & \underline{{Major/Conc.}} & \underline{Degree, Year} \\
Undergraduate: & Northern Arizona State University & Physics/Mathematics & B.S., 1993 \\
Graduate: & Arizona State University & Physics & Ph.D., 2001 \\
Postdoctoral: & Arizona State University & Physics & 2002-2006 
\end{tabular}
\end{table}

\noindent{\bf{Appointments}}
\begin{table}[ht]
%\vspace{2mm}
\centering
\begin{tabular}{lll}
Associate Professor (tenure-track) & Arizona State University &2017 to present\\
Associate Research Professor & Arizona State University & 2013 to 2017\\
Assistant Research Professor & Arizona State University & 2006-2013 \\
Postdoctoral Research Associate & Arizona State University &2002-2006\\
\end{tabular}
\end{table}

%\noindent \textbf{Invited talks over the last three years} \newline \newline
%``Overview of Spectroscopy Results in Meson Photoproduction with Polarization Observables,''
%HADRON, Newport News, VA., Septrmber 14, 2015.
%\newline

\noindent{\bf Synergistic activities}
\vskip0.15in
GlueX Collaboration Board member, 2013-2015;
Division of Nuclear Physics, Conference Experience for Undergraduates, 
Review Committee member, 2018;
International Advisory Committee, MENU Conference, 2018-Present;
%CLAS Collaboration, Analysis Coordinator for $\Xi$ and $\Omega^-$ photoproduction, 
%2017-Present.
\newline

\begin{bibunit}
\makeatletter 
\renewcommand\@biblabel[1]{#1.}
\makeatother
\makeatletter
\renewenvironment{thebibliography}[1]
     {\list{\@biblabel{\@arabic\c@enumiv}}%
           {\settowidth\labelwidth{\@biblabel{#1}}%
            \leftmargin\labelwidth
            \advance\leftmargin\labelsep
            \@openbib@code
            \usecounter{enumiv}%
            \let\p@enumiv\@empty
            \renewcommand\theenumiv{\@arabic\c@enumiv}}%
      \sloppy
      \clubpenalty4000
      \@clubpenalty \clubpenalty
      \widowpenalty4000%
      \sfcode`\.\@m}
     {\def\@noitemerr
       {\@latex@warning{Empty `thebibliography' environment}}%
      \endlist}
\makeatother
\noindent{
\textbf{Publications over the last three years}
\phantom{~\cite{
CLAS:2021udy,CLAS:2021osv,Hayward:2021psm,CLAS:2020ngl,
PhysRevC.103.L022201,
CLAS:2020vpr,
ADHIKARI2021164807,
CLAS:2020rdz,
PhysRevC.101.065206,
CLAS:2020mom,
PhysRevC.100.052201,
PhysRevLett.123.072001,
CLAS:2018mmb,CLAS:2018kvn,CLAS:2018azo}
%Kunkel:2017src,Roy:2017qwv,Akbar:2017uuk,
%Anisovich:2017pox,Collins:2017vev,Mattione:2017fxc,
%Anisovich:2017rpe,Dugger:2017zoq,Collins:2017sgu,AlGhoul:2017nbp,
%Paterson:2016vmc,Ghoul:2015ifw} 
as shown below.}}
\putbib[bibtexrefs2021]
\end{bibunit}

